Title: Housing equity for health equity--A rights-based approach to the control of Lassa fever in post-war Sierra Leone

Authors:

J. Daniel Kelly (jdkelly@bcm.edu)
M. Bailor Barrie (bailor@wellbodyalliance.org)
Rachel A. Ross (rross4@partners.org)
Brian A. Temple (brian.temple@ttuhsc.edu)
Lina M. Moses (lmoses2@tulane.edu)
Daniel G. Bausch (dbaush@tulane.edu)

Version: 2 Date: 15 November 2012

Author's response to reviews: see over